Author's response to reviews

Title: The impact of smoking on adherence to treatment for latent tuberculosis infection

Authors:
- Melanie Lavigne (mel_lavigne@hotmail.com)
- Isabelle Rocher (isabelle.rocher@inbpq.qc.ca)
- Colin Steensma (colin.steensma@phac-aspc.gc.ca)
- Paul Brassard (paul.brassard@mcgill.ca)

Version: 3 Date: 31 January 2006

Author's response to reviews:

Dear Editor

We revised the manuscript as requested and included a statement regarding informed consent in the last sentence of the first paragraph of the methods section on page 3:

- A written informed consent to participate in the study was obtained before administration of the questionnaire.

We also made sure that the format of our manuscript conforms with your requirements.

We certainly hope you will find our manuscript suitable for publication in BMC Public Health.

Thank you for your collaboration.